Minutes ~ RSC Weekly Meeting During RUN03 ~ April 4, 2003

(1) Status of RHIC Commissioning [Haixin]

    - the beams have been injected and captured in both rings

    - the tuning at injection is done

    - PLL commissioning is still underway, so presently being used without feedback.

    - the tuning of ramp has 90% efficiency after 2-4 days of tuning (most of the loss is during the beta* squeeze at the end of the ramp) and they are still working to improve it.  For spin, the tune looks pretty good, though it might be a bit close to the 3/14 resonance in a few spots.  This will be monitored.

    - plan is to get 55x55 bunches going and try 110x110 this weekend;

      then look at polarization and start to do work on commissioning the spin rotators and spin flipper.

(2) Status AGS Commissioning [report by Leif]

    - no polarized beam 'til Friday because of the death of a tube in the source.  The tube has been replaced with a spare and was operational at the time of the meeting.

    - without polarization, the focus was on the reducing the growth of the emittance during the ramp; n.b., smaller emittance is good for polarized protons.  Scrapping is effective longitudinally and vertically, yet not so much horizontally.

    - there are two or three machine setup changes which can be explored to improve polarization retention.  

(3) Status of AGS CNI Polarimeter [report by Sandro]

    - the polarimeter is operation and doing great; it has been crossed check with the E880 polarimeter.

    - as anticipated, the silicon detectors have received about 10 x (last year's RHIC) radiation exposure and are becoming noisy.  Sandro proposes that these be replaced on this Wednesday.  

    - use of AGS CNI on the ramp is still being developed.

    - there was a discussion about the intensity limit. At higher intensities or unstable beam, the electronics start to oscillate due to what is believed to be prompt background.  Effects are consistent with saturation of preamps, so, on Wednesday, they will modify preampliers.  Others suggested putting in some scintillators in the vicinity of the "best guess" for the source of the background to try to help isolate it and then have the tune adjusted to avoid the problem.

(4) Status of RHIC CNI Polarimeters [report by Dima]

    - the hardware is all ready to take data w/120 bunch mode, in both 

      histogram (as last year) and event mode (as w/this year's AGS CNI).  In event mode, they can collect 1.2M/strip.

    - the software from last year is able to process the data if the data is collected as histograms like last year.  He is working on porting the code developed for the AGS CNI to RHIC so that data can be collected and analysed on the event level.  The readout and analysis may take something like 5 minutes from the end of data collection in this mode.

(5) Plans for Commissioning the Spin Rotators [discussion, led by Waldo]

    - the rotator magnets have been checked 

    - the BPMs have been checked and, though some are not wired right, Waldo knows what their wiring is and can compensate for it.

    - the first order of business will be to turn them on while the beam is at full energy.  Waldo wants to do this slowly as was done with the snakes so that the orbit can be corrected.  

    - for local polarimetry, PHENIX will measure neutron asymmetries using the SMD+ZDC (one hour or so at 40% polarization, ~10 luminosity of Run2) and STAR will use the BBC (fast) and the forward pion detectors (not so fast).  

(6) Plans for Commissioning the Spin Flipper [discussion, led by Mei]

    - setup of controls still underway for operator control of the device; the operational plan for it (at the moment) would be:

       (1) announce on PA that spin flip will start in 2 minutes

       (2) all experiments stop taking data when they hear the announcement

       (3) an event is broadcast via the V102 module to say that the spin flip is underway.  This signal should be latched into the experiments DAQ system so that, if the experiment does not turn off, then the data is marked as "no good".

       (4) measure the polarization in both rings using the RHIC CNI 

           polarimeters (takes about 2 minutes all told)

       (5) detune the snake in the ring for which the spin will be flipped (will about 5 minutes due to need for slow ramp rate in snakes)

       (6) flip the spin

       (7) restore the tune of the snake (another 5 minutes)

       (8) measure the polarization in both rings

       (9) update the CDEV database & V124 modules with the new spin pattern

      (10) increment the "spin flip" counter maintained in CDEV for the ring which was flipped

      (11) an event is broadcast via the V102 modules to say that the spin flip is done

      (12) announce on PA that spin flip is done

    - though this "complete" system is not ready yet, Mei can do things in expert mode to commission the device once we have polarized beam in RHIC (even at injection energies) & the RHIC CNI polarimeters are operational.  We favor early work on the device during the commissioning period because, in the limited time available to her in Run-2, the device showed promise.

(7) Plans for the Spin Pattern [start of discussion for the 4/11 meeting]

    - As he did last year, Waldo agreed to come up with a proposed spin

      pattern for the 4/11 meeting.  

    - Gerry suggested that we consider skipping the known "bad" bunches, bunch 1 (bad for unknown reason) and bunch 11 (the one which is tapped by the tune measurement system).  In particular, the unpolarized bunch should not be one of the bad bunches.  

    - One other thaught was whether or not the "tapped" bunch can be one of the bunches which is bad.  

    - We should discuss this issue on 4/11.

(8) Plans for Identifying "Good/Bad Fill" and "End of Store" Criteria

    - Since time is precious, we would rather not keep a fill for 12 hours (or even one) if the fill is bad.   In a lot of cases, the RHIC polarimeter can identify a bad fill, quite quickly and we would just want to dump it and retry. 

    - Brendan proposed that we have a group of people representing the 

      various efforts come up with the criteria.  He proposed that this

      group should include Osamu Jinnouchi (RHIC polarimeter), Joanna Kiryluk (STAR), Misashi Kaneta (PHENIX) w/Yuji or Brendan since Misashi was not part of the PHENIX Run-2 effort and may need some help with the history, and one machine person to be named later.  

    - We should discuss this issue on 4/11.

